Author's response to reviews

Title: Effects of Lactobacillus salivarius-containing tablets on caries risk factors: open-label clinical trial

Authors:

Tetsuyo Nishihara (tetsuyo.oyustet@gmail.com)
Nao Suzuki (naojsz@college.fdcnet.ac.jp)
Masahiro Yoneda (yoneda@college.fdcnet.ac.jp)
Takao Hirofuji (hirofuji@college.fdcnet.ac.jp)

Version: 2 Date: 10 June 2014

Author's response to reviews:

I revised the page numbers of the text described in the cover letter. In addition, figures 1A and 1B were improved and uploaded.